Reviewer 1

This study concludes the risk of bleeding in PSPH by Gastroscopy grade and S / C ratio. The content is generally agreed, but there is one contradiction. I will write it together with other matters to be added.

Q1: In conclusion, there is a discrepancy in the title because CT and endoscopic results are required. You should change the title.

A1: The reviewing professor's opinion is correct. In fact, we should use CT combined with gastroscopy to diagnose PSPH. For this we have modified the title: Computed Tomography Combined with Gastroscope Application in Pancreatic Segmental Portal Hypertension. Thank you for your suggestion!

Q2: In this study, the results of CT are mainly emphasized, but the CT imaging method and contrast agent usage are not clarified. The CT imaging method should be specifically stated.

A2: We are very sorry that we neglected to explain the CT imaging modalities. We have supplemented this in the article (Materials and methods-Method-Computed Tomography Scanning Solution). Thanks again for your tips!

Q3: The S/C ratio plays an important role, so I think a figure of the measurement method is necessary.

A3: We agree with the reviewing professor's point of view. We modified Fig.4 to add 'C. Splenic portal dilated splenic vein' and enlarged the image of the measurement method.

Q4: The Gastorography grade column in Table 3 is blank and only the P value is shown. I don't understand this.

A4: Thank you very much for reminding. We have increased the number and proportion of G1, G2, and G3 patients based on your comments.
Reviewer 2

Thank you for giving me an opportunity to review the article. My suggestions:

Q1: Great deal of English polishing is needed.
A1: We are very grateful to the reviewing professors for the initial revisions to our language. At the same time, we asked AJE’s language editor to polish our manuscript and upload the language certificate.

Q2: The results are difficult to understand and clinical relevance is questionable.
A2: Studies with small samples have certain limitations. We added the limitations before the conclusion.

Q3: A tabular presentation of data and its literature is needed in discussion.
A3: Thanks to the reviewers for the reminder. We present the relevant literature and data in the Discussion section.

Q4: Please mention all the study limitations in separate paragraph before conclusion.
A4: As in the previous question, we've added limitation part.

Q5: Please mention which software was used for statistics.
A5: The statistical software and methods used are described in detail in the Methods section.

The editor and two reviewing professors were very helpful in our research.

We sincerely thank you for this!